

Author's response to reviews

Title: Establishing confidence in the output of qualitative research synthesis: the ConQual approach

Authors:

Zachary Munn (Zachary.Munn@adelaide.edu.au)

Kylie Porritt (kylie.porritt@adelaide.edu.au)

Craig Lockwood (craig.lockwood@adelaide.edu.au)

Edoardo Aromataris (edoardo.aromataris@adelaide.edu.au)

Alan Pearson (alan.pearson@adelaide.edu.au)

Version:7 **Date:**25 August 2014

Author's response to reviews:

Dear Editorial Team,

We have made the revisions requested and have included a 'competing interests' section. There are no competing interests to declare.

Kind Regards,

The Authors